UNIVERSITY BIRMINGHAM University of Birmingham Research at Birmingham

Chromosomal polymorphisms in assisted reproduction

Ralapanawe, Madara S B; Gajaweera, Sugandika Lakmali; Karunaratne, Nishendra; Price, Malcolm James; Melo, Pedro; Coomarasamy, Arri; Gallos, Ioannis

DOI: 10.1530/RAF-21-0116

License: Creative Commons: Attribution-NonCommercial-NoDerivs (CC BY-NC-ND)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Ralapanawe, MSB, Gajaweera, SL, Karunaratne, N, Price, MJ, Melo, P, Coomarasamy, A & Gallos, I 2022, 'Chromosomal polymorphisms in assisted reproduction: an analysis of 942 cycles', *Reproduction and Fertility*, vol. 3, no. 3, pp. 133-139. https://doi.org/10.1530/RAF-21-0116

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

133-139

Chromosomal polymorphisms in assisted reproduction: an analysis of 942 cycles

Madara S B Ralapanawe^{1,2}, Sugandika Lakmali Gajaweera², Nishendra Karunaratne², Malcolm James Price^{3,4}, Pedro Melo¹, Arri Coomarasamy¹ and Ioannis Gallos¹

¹Tommy's National Centre for Miscarriage Research, Institute of Metabolism and Systems Research, University of Birmingham, Birmingham, UK ²Fertility Centre, Lanka Hospitals Corporation Plc, Colombo, Sri Lanka

³Institute of Applied Health Research, University of Birmingham, Birmingham, UK

⁴NIHR Birmingham Biomedical Research Centre, University Hospitals Birmingham NHS Foundation Trust and University of Birmingham, Birmingham, UK

Correspondence should be addressed to M S B Ralapanawe: madara.ralapanawe@gmail.com

Abstract

The use of intracytoplasmic sperm injection (ICSI) has recently increased worldwide. The live birth rate per ICSI cycle is low, and over half of infertile couples remain childless. Chromosomal polymorphisms are up to five times more common in couples with infertility compared to the general population. We aimed to investigate the association between chromosomal polymorphisms and reproductive outcomes in couples undergoing ICSI treatment. We analysed 942 ICSI fresh and frozen embryo transfer cycles in 697 women who underwent karyotyping analysis using Giemsa-Trypsin-Leishman banding prior to assisted conception at the Fertility Centre of Lanka Hospitals, Sri Lanka, between 2016 and 2018. The primary outcomes were pregnancy, miscarriage, and live birth rates. We compared outcomes according to the presence or absence of chromosomal polymorphism in females, males and couples. There were 294 pregnancies (31.2%) recorded in the study; 130 suffered a miscarriage (13.8%), 13 were ectopic pregnancies (1.3%) and 151 resulted in a live birth (16.0%). The evidence from univariable and multivariable analyses (adjusted for age, BMI, ovarian reserve and treatment type) did not confidently identify a difference in pregnancy, miscarriage or live birth rates between couples with no chromosomal polymorphisms compared to couples where the female, male or both partners were carriers of a chromosomal polymorphisms. Further, we did not identify a clear association between the presence of chromosomal polymorphisms and reproductive outcomes compared to participants without chromosomal polymorphisms. Wide Cls precluded the identification of clinically meaningful associations.

Lay summary

Infertility affects approximately one in eight couples worldwide. The use of intracytoplasmic sperm injection (ICSI), where the sperm is directly injected into an egg using a micromanipulator outside the body, has become particularly popular in recent years. However, the success rate remains low. In human cells, the genetic material is arranged in structures called chromosomes. Chromosomal polymorphism is a normal variation where the genetic material is arranged differently to the average individual and is more common in infertile couples compared to the general population. We analysed data from 942 ICSI cycles in 697 couples who underwent karyotyping analysis to assess the changes in chromosomes between 2016 and 2018. The pregnancy rate was 31.2%, with 16.0% of participants experiencing a live birth, while 13.8% of pregnancies resulted in a miscarriage and 1.3% were outside the womb cavity (ectopic). The evidence did not identify a clear association between the chromosomal polymorphism and the outcome of treatment.

Key Words: • infertility • assisted reproductive treatment

chromosomal polymorphism
 pregnancy outcomes

Reproduction and Fertility (2022) 3 133-139

https://raf.bioscientifica.com https://doi.org/10.1530/RAF-21-0116 © 2022 The authors Published by Bioscientifica Ltd



This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License.

Introduction

Infertility is common, affecting one in eight couples worldwide (ESHRE 2020). Assisted reproductive technology (ART), including in vitro fertilisation (IVF) and intracytoplasmic sperm injection (ICSI), is the mainstay treatment for couples with infertility. More than two million ART cycles are performed worldwide every year, and this number is steadily increasing (Mascarenhas et al. 2012, ESHRE 2022). The use of ICSI has become particularly popular in recent years, with double the number of cycles globally compared to conventional IVF (ESHRE 2018). However, the live birth rate per ICSI cycle remains relatively low (~30%) (HFEA 2016).

Chromosomal polymorphisms are normal variations that occur in 2-5% of the general population. They are usually found in the genetically inactive heterochromatic regions of chromosomes (Wyandt et al. 2017) and have no clear impact on phenotype (Brothman et al. 2006), although in many species chromosomal polymorphisms result in reduced fertility (Kirkpatrick et al. 2010). In humans, polymorphisms are also up to five times more common in couples with infertility compared to the general population (Xu et al. 2016). The presence of polymorphism affects spermatogenesis adversely and could be detrimental to the outcome of ICSI (Nakamura et al. 2001, Yakin et al. 2005). In addition, increased rates of recurrent miscarriages and other adverse obstetric outcomes have been associated with chromosomal polymorphism (Minocherhomji et al. 2009, Ahmet Okay et al. 2010, Pokale 2015).

Chromosomal polymorphism is described as the presence of variants in the heterochromatin region of the chromosome. The constitutive heterochromatin is the stable form present in the polymorphic variants. An increase or decrease of the heterochromatin region of the long arm of the chromosome constitutes heterochromatic segments (non-acrocentric). Chromosomal polymorphism may also manifest through increases in the length of the short arm of the chromosome (acrocentric) with a satellite stalk, satellite or a double satellite. Finally, the International System for Human Cytogenetic Nomenclature considers gene inversions [Inv(9)] to also fall within the definition of chromosomal polymorphism (Shaffer *et al.* 2013).

A recent systematic review of chromosomal polymorphism in assisted reproduction found an association with higher rates of miscarriage which was sexdependent given the higher miscarriage rate observed in female carriers of chromosomal polymorphism compared to male carriers (Ralapanawe *et al.* 2022). The review did not find evidence related to chromosomal polymorphisms having any adverse effects on rates of pregnancy, clinical pregnancy, ongoing pregnancy, preterm birth and live birth after IVF or ICSI, irrespective of whether the carrier was the female partner, the male partner or both. In addition, the systematic review called for further research to confirm the association between polymorphic variations in females and miscarriage and to strengthen the certainty of the evidence for other reproductive outcomes. We propose that if miscarriage rates are higher in patients with chromosomal polymorphism, it is reasonable to hypothesise that there could be a knock-on effect on other pregnancy outcomes. Here we explore the association between chromosomal polymorphisms and reproductive outcomes in couples undergoing ICSI treatment.

Materials and methods

Study design

This study retrospectively investigated couples undergoing karyotyping analysis followed by a cycle of ICSI treatment at the Fertility Centre of Lanka Hospitals Corporation Plc, Sri Lanka, from January 2016 to December 2018. Pregnancy outcomes were collected until November 2019.

We excluded couples undergoing treatment with donor gametes, with numerical or structural abnormalities in karyotyping or absence of karyotyping reports, poor follicular development, abnormal cleavage or blastocyst formation, freeze-all cycles and records where pregnancy outcomes had not been documented.

Karyotype analysis

Karyotyping was performed on peripheral blood leukocytes. The standard laboratory protocol using Giemsa-Trypsin-Leishman banding was followed for all samples. Twenty metaphases were counted and analysed. Four to five karyotypes were analysed at a banding resolution of 550×. The karyotyping results were reviewed by two analysts independently.

Ovarian stimulation, ICSI and embryo culture

All female participants were stimulated with a long protocol using GnRH agonist 0.1 mg (triptorelin/Decapeptyl, Ferring GmbH, Wittland, Germany) combined with recombinant follicle-stimulating hormone (FSH) 150–450 IU (follitropin alfa/Gonal-f, Merck Serono, Modugno (BA), Italy) or a short protocol with GnRH antagonist 0.25 mg (cetrorelix

3:3

135

acetate/Cetrotide, Baxter Oncology GmbH, Halle, Germany) combined with recombinant FSH 150-450 IU. After the evaluation of serum oestradiol level (1000-5000 pg/mL) on the tenth day, recombinant human chorionic gonadotrophin (hCG) 250 µg (choriogonadotropin alfa/ Ovidrel, Merck, Serono S.p.A., Modugno (BA), Italy) was administered subcutaneously. Oocyte recovery was performed 35 h after the hCG injection. Following oocyte insemination with ICSI, embryos were cultured (Vitrolife Sweden AB, V.Frolunda, Sweden) for up to 3 days. All embryos with more than six cells were selected. Two embryos were transferred per fresh cycle, and the remaining embryos were vitrified. In women where a fresh transfer was not possible, we performed cryopreservation of all embryos and carried out frozen embryo transfer (FET) at a later date.

Embryo transfer

Two cleavage-stage fresh embryos were transferred per cycle, and the remaining embryos were vitrified. Subsequent FET cycles involved warming and transfer at cleavage stage (six to eight cells) or further culture of embryos for 2 days until blastocyst formation.

Outcomes and follow-up

Pregnancy was confirmed 2 weeks after embryo transfer (Serum β HCG >10 mIU/mL). The primary outcomes included pregnancy rate (gestational age 4-6 weeks), miscarriage rate (gestational age less than 12 weeks) and live birth rate (gestational age over 32 weeks). Outcome data were analysed per female, male and couple according to the presence or absence of chromosomal polymorphism. There were no missing data for demographic characteristics including age, BMI, FSH, luteinising hormone (LH), thyroid-stimulating hormone (TSH), free thyroxine (T4) and prolactin. The pregnancy rate refers to positive pregnancies for the cycles with embryo transfers. Miscarriage refers to pregnancy losses calculated from the total number of treatment cycles. Live birth rate refers to the total number of live babies from the total number of fresh and frozen embryo transfers.

Statistical analysis

Baseline characteristics and outcome data were described with proportions for binary data or means with S.D. or median and interquartile range for continuous variables, as appropriate. The rates of the reproductive outcomes were plotted graphically using proportions and 95% CIs. Complete case analysis was adopted. Logistic regression models were fitted to estimate the unadjusted and adjusted odds ratio for confounding variables including age, BMI, FSH, LH and type of treatment (fresh vs frozen). All statistical analyses were done using Stata Statistical Software (Release 16, TX, USA).

Ethical consideration

The ethics committee of Lanka Hospitals Corporation PLC granted permission for the use of the patient record data database following the review of the study protocol.

Results

There were 1879 ICSI and FET cycles performed at the Fertility Centre during the study period. In total, 937 fresh ICSI and FET cycles were excluded from the analysis due to the use of donor gametes, absence of karyotyping reports, numerical and structural abnormalities in karyotyping, poor follicular development, abnormal cleavage and blastocyst formation, embryo vitrification without subsequent transfer and records without pregnancy outcomes. Figure 1 shows the data selection process. There were 149 participants who underwent long (n = 114) or short (n = 35) protocol stimulation and did not proceed with FET due to hyperstimulation or any other factors but went on to have FET at a later date. In total, 942 treatment cycles (548 fresh ICSI cycles and 394 FET cycles) from 697 couples were included in the study.

Table 1 contains baseline characteristics of the studypopulation.SupplementaryTable 1 (see section on

Total number of ICSI fresh and FET cycles recorded from 2016 to 2018 (n = 1879)
Total number of cycles (n = 1879)
Cycles excluded (n = 937)
- Using donor oocytes (n = 375)
- Using donor semen (n = 31)
 Absence of karyotyping reports (n = 201)
 Numerical and structural abnormalities in karyotyping (n = 13)
 Poor follicular development (n = 29)
 Abnormal embryo cleavage (no transfer) (n = 54)
 Abnormal blastocyst formation (no transfer) (n = 4)
 All embryos vitrified for FET transfers (n = 149)
 No records of pregnancy outcome (n = 81)
Total number of fresh ICSI and FET cycles studying the effect of chromosomal

Total number of fresh ICSI and FET cycles studying the effect of chromosomal polymorphism n = 942 (Total number of couples n = 697)

Figure 1 Flow chart of data selection process.



This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License.

ND aded from Bioscientifica.com at 03/22/2023 11:25:31AM via University of Birmingham and PERPETUAL ACCESS BIRMINGHAM **Table 1**Baseline characteristics of the study population(*n* = 942). Data are presented as *n* (%) or as mean ± S.D.

Characteristics	Values
Age	34 ± 4.1
BMI	24 ± 3.8
FSH	6.5 ± 1.8
LH	5.8 ± 2.7
Treatment type	
ICSI cycles	
Long agonist	407 (43.2)
Short antagonist	141 (15)
FET cycles	
Cleavage stage transfers (day 3)	219 (23.2)
Blastocyst stage transfers (day 5)	175 (18.6)
Oocytes retrieved	15.5 ± 8.2
Mature oocytes	15.0 ± 8.2
Fertilised oocytes	11.2 ± 7.4
Cleavage embryos (day 3)	7.5 ± 5.1
Chromosomal polymorphism	
Females with polymorphism	150 (15.9)
Males with polymorphism	200 (21.2)
Couples with polymorphism	144 (15.3)
Couples without polymorphism	448 (47.6)

FET, frozen embryo transfer; FSH, follicle-stimulating hormone; ICSI, intracytoplasmic sperm injection; LH, luteinising hormone.

supplementary materials given at the end of this article) contains types of chromosomal polymorphic variants present in female and male participants in the study population.

From the 942 cycles analysed, in 144, both partners were carriers of polymorphisms (15.3%); in 150, only the female partners were carriers of polymorphisms (15.9%); in 200, only the males were carriers of polymorphisms (21.2%); and in 448 cycles, neither partner carried a polymorphism (47.6%).

There were 294 pregnancies (overall pregnancy rate 31.2%; ICSI pregnancy rate 24.3% (133/548); FET 40.9% (161/394) recorded in 942 cycles in the study); of which, 130 suffered a miscarriage (overall miscarriage rate 13.8%; ICSI miscarriage rate 11.3% (62/548); FET 17.2% (68/394)), 13 had an ectopic pregnancy (1.3%; ICSI 1.5% (8/548);

FET 1.3% (5/394)) and 151 had a live birth (overall live birth rate 16.0%; ICSI live birth rate 11.5% (63/548); FET 22.3% (88/394)). The total number of participants with chromosomal polymorphic variants was 494 (52.4%), while 448 (47.6%) did not exhibit any of the polymorphic variants. Table 2 shows details of pregnancy, miscarriage and live birth rates according to the presence or absence of chromosomal polymorphism.

Table 3 presents the unadjusted and adjusted odds ratios for factors influencing the rates of pregnancy, miscarriage and live birth. We found no association between chromosomal polymorphisms and these reproductive outcomes.

Figure 2 shows the point effect estimates and respective CIs for outcomes of pregnancy, miscarriage and live birth for the whole cohort and for females, males and couples with polymorphism.

Discussion

In this analysis, we found no evidence of a difference in pregnancy, miscarriage or live birth rates between participants without polymorphisms and in those where one or both partners were carriers of chromosomal polymorphisms. This was observed in the unadjusted univariate analysis and multivariate analysis adjusted for age, BMI, ovarian reserve markers and treatment type. Although some of our point estimates suggest a clinically important impact, the CIs were wide and crossed the line of no effect.

In this study, some participants did not proceed with FET due to hyperstimulation or other factors and underwent FET instead. A small proportion of outcome data on pregnancy, miscarriage and live birth were missing or not reported and were not included in the study. The study sample was large, but we cannot rule out a type II error. The attrition or loss to follow-up rate were low, and we were able to adjust the result for potential confounders.

Table 2 Pregnancy, miscarriage and live birth rates of carriers and non-carriers of chromosomal polymorphism.

Polymorphism	n	Pregnancy rate (%)	Miscarriage rate (%)	Live birth rate (%)
	40.4	150 (21.0)	72 (14.0)	
Females, males or couples with polymorphism	494	156 (31.6)	73 (14.8)	79 (16.0)
Females with polymorphism	150	36 (24)	19 (12.7)	16 (10.7)
Males with polymorphism	200	68 (34)	28 (14)	38 (19)
Couples with polymorphism	144	52 (36.1)	26 (18.1)	25 (17.4)
Couples without polymorphism	448	138 (30.8)	57 (12.7)	72 (16.1)
Total	942	294 (31.2)	130 (13.8)	151 (16.0)

Ectopic pregnancies (n = 13, 1.3%) were excluded from the miscarriages



This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License.

ND aded from Bioscientifica.com at 03/22/2023 11:25:31AM via University of Birmingham and PERPETUAL ACCESS BIRMINGHAM

Outcome	Unadjusted O	R	Adjusted OR	
	OR (95% CI)	P-value	OR (95% CI)	P-value
Pregnancy				
Females, males or couples with polymorphism	1.03 (0.78–1.36)	0.79	1.05 (0.79–1.39)	0.73
Females with polymorphism	0.70 (0.46–1.08)	0.11	0.70 (0.45–1.08)	0.10
Males with polymorphism	1.15 (0.81–1.64)	0.42	1.19 (0.82–1.71)	0.34
Couples with polymorphism	1.26 (0.85–1.88)	0.23	1.29 (0.86–1.93)	0.21
Miscarriage				
Females, males or couples with polymorphism	1.18 (0.81–1.72)	0.36	1.20 (0.83–1.73)	0.32
Females with polymorphism	0.99 (0.57–1.73)	0.98	1.00 (0.57–1.75)	0.99
Males with polymorphism	1.11 (0.68–1.81)	0.65	1.13 (0.69–1.86)	0.60
Couples with polymorphism	1.51 (0.90–2.51)	0.11	1.54 (0.92–2.57)	0.09
Live birth				
Females, males or couples with polymorphism	0.99 (0.70-1.40)	0.97	1.00 (0.70-1.44)	0.95
Females with polymorphism	0.62 (0.35–1.10)	0.10	0.61 (0.34–1.10)	0.10
Males with polymorphism	1.22 (0.79–1.89)	0.35	1.28 (0.82–2.00)	0.27
Couples with polymorphism	1.09 (0.66–1.80)	0.71	1.10 (0.65–1.83)	0.71

 Table 3
 Unadjusted and adjusted odds ratio for pregnancy, miscarriage and live birth rates.

The reference category is no chromosomal polymorphism in either partner. OR, odds ratio.

Our findings are consistent with existing literature summarised in our previous systematic review of observational studies (Ralapanawe et al. 2022). The review suggested that there was a paucity of evidence on whether polymorphic variation in individuals (males or females) or couples adversely affects the chance of a pregnancy, miscarriage and live birth following ICSI, except for miscarriage in the presence of chromosomal polymorphism in females. However, nine studies in the systematic review involved participants of Chinese origin, and extrapolation to other cohorts may not be appropriate.

The existing literature is conflicting, with some authors reporting that chromosomal polymorphisms are associated with adverse reproductive outcomes (Xiaobin et al. 2012, Cheng et al. 2017), while others have identified no such association (Hong et al. 2011, Liang et al. 2014, Song et al. 2017). It is possible that our study may have been underpowered to detect any differences. Further, a small adverse effect of chromosomal polymorphisms upon reproductive outcomes may exist for some populations but not others. There is a need for additional prospective studies evaluating the association between chromosomal polymorphisms and reproductive outcomes in patients of multiple ethnicities.

Finally, future research should investigate whether there is an adverse effect from specific high-risk chromosomal polymorphisms on reproductive outcomes. There is evidence that specific types of polymorphisms including non-acrocentric and Yqh in male patients may exhibit a particularly strong association with reproductive outcomes (Yakin et al. 2005, Sipek Jr et al. 2014, Xu et al.



Figure 2 Confidence intervals of unadjusted and adjusted odds ratios of pregnancy, miscarriage, and live birth of female, male and couple with chromosomal polymorphism.



This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License ND aded from Bioscientifica.com at 03/22/2023 11:25:31AM

via University of Birmingham and PERPETUAL ACCESS BIRMINGHAM

3:3

2016). It remains unclear, however, whether these highrisk polymorphisms are associated with adverse outcomes following ART.

Supplementary materials

This is linked to the online version of the paper at https://doi.org/10.1530/ RAF-21-0116.

Declaration of interest

The authors declare that there is no conflict of interest that could be perceived as prejudicing the impartialities of the research reported.

Funding

This study did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

Author contribution statement

M S B R and I G were responsible for defining the research question; M S B R, S L G and I G conceptualised and designed the work; M S B R and S L G were responsible for the data acquisition; M J P, M S B R, S L G, and I G performed the statistical analysis; S L G assisted in the design of the figures and tables, and in the manuscript preparation; All authors including N K assisted with interpretation of the findings; M S B R drafted the manuscript. P M and I G revised the final manuscript. All authors revised the manuscript critically for important intellectual content and gave final approval of the version to be published. A C is the guarantor.

Acknowledgements

The authors would like to acknowledge all the staff of Lanka Hospitals Fertility Centre for maintaining the clinical database and especially Ridma Chandani, Madu Sammani, Thilini Pabasara, Amilka Lasanthi, Himashi Hapangama, Gimhani Weerakkody, Menakadevi and Yogaletchami for their support to search and retrieve clinical data for the analysis. M J P was supported by the NIHR Birmingham Biomedical Research Centre at the University Hospitals Birmingham NHS Foundation Trust and the University of Birmingham. The views expressed in this publication are those of the authors and not necessarily those of the NIHR, NHS or the UK Department of Health and Social Care. The authors would also like to express their gratitude to Dr Sayeed Haque, Senior lecturer of medical statistics, Institute of Applied Health, University of Birmingham for helping with the CI plots.

References

Ahmet Okay C, Isilay O, Fatma D & Munis D 2010 Cytogenetic results of patients with infertility in middle Anatolia, Turkey: do heterochromatin polymorphisms affect fertility? *Journal of* *Reproduction and Infertility* **11** 179–181. (available at: https://pubmed. ncbi.nlm.nih.gov/23926487/)

- Brothman AR, Schneider NR, Sajkevych I, Cooley LD, Butler MG, Patil S, Mascarello JT, Rao KW, Dewald GW, Park JP, *et al.* 2006 Cytogenetic heteromorphisms; survey results and reporting practices of Giemsa-band regions that we have pondered for years. *Archives of Pathology and Laboratory Medicine* **130** 947–949. (https://doi. org/10.5858/2006-130-947-CHSRAR)
- Cheng R, Ma Y, Nie Y, Qiao X, Yang Z, Zeng R & Xu L 2017 Chromosomal polymorphisms are associated with female infertility and adverse reproductive outcomes after infertility treatment: a 7-year retrospective study. *Reproductive Biomedicine Online* **35** 72–80. (https:// doi.org/10.1016/j.rbmo.2017.03.022)
- European Society of Human Reproduction and Embryology 2018 European Society of Human Reproduction and Embryology. In 34th Annual Meeting, Barcelona, Spain. (available at https://www.eshre.eu/Annual-Meeting/Barcelona-2018)
- European Society of Human Reproduction and Embryology 2020 ART Fact Sheet. ESHRE.
- European Society of Human Reproduction and Embryology 2022 ART Fact Sheet. ESHRE. (available at: https://www.eshre.eu/Press-Room/ Resources)
- Hong Y, Zhou YW, Tao J, Wang SX & Zhao XM 2011 Do polymorphic variants of chromosomes affect the outcome of in vitro fertilisation and embryo transfer treatment? *Human Reproduction* 26 933–940. (https://doi.org/10.1093/humrep/deq333)
- Human Fertilisation and Embryology Authority 2016 Fertilitytreatment-2014; trends and figures, version 1. (available at https:// www.hfea.gov.uk/media/1783/fertility-treatment-2014-trends-andfigures.pdf)
- Kirkpatrick M 2010 How and why chromosome inversions evolve. PLoS Biology 8 e1000501. (https://doi.org/10.1371/journal.pbio.1000501)
- Liang J, Zhang Y, Yu Y, Sun W, Jing J & Liu R 2014 Effect of chromosomal polymorphisms of different genders on fertilization rate of fresh IVF-ICSI embryo transfer cycles. *Reproductive Biomedicine Online* 29 436–444. (https://doi.org/10.1016/j.rbmo.2014.06.011)
- Mascarenhas MN, Flaxman SR, Boerma T, Vanderpoel S & Stevens GA 2012 National, regional, and global trends in infertility prevalence since 1990: a systematic analysis of 277 health surveys. *PLoS Medicine* **9** e1001356. (https://doi.org/10.1371/journal.pmed.1001356)
- Minocherhomji S, Athalye AS, Madon PF, Kulkarni D, Uttamchandani SA & Parikh FR 2009 A case-control study identifying chromosomal polymorphic variations as forms of epigenetic alterations associated with the infertility phenotype. *Fertility and Sterility* **92** 88–95. (https://doi.org/10.1016/j. fertnstert.2008.05.071)
- Nakamura Y, Kitamura M, Nishimura K, Koga M, Kondoh N, Takeyama M, Matsumiya K & Okuyama A 2001 Chromosomal variants among 1970 infertile men. *International Journal of Urology* 8 49–52. (https://doi.org/10.1046/j.1442-2042.2001.00242.x)
- **Pokale VS** 2015 Does a heterochromatic variant affect the human reproductive outcome? Research. *Journal of Recent Sciences* **4** 108–113.
- Ralapanawe MSB, Khattak H, Hapangama HR, Weerakkody GR, Papadopoulou A & Gallos I 2022 Chromosomal polymorphisms in assisted reproduction: a systematic review and meta-analysis. *Human Fertility* In Press. (https://doi.org/10.1080/14647273.2022.2051614)
- Shaffer LG, McGovan-Jordan L & Schmid M (eds) 2013 ISCN 2013: An International System for Human Cytogenetic Nomenclature. Karger Medical and Scientific Publishers.
- Sipek Jr A, Mihalova R, Panczak A, Hrckova L, Janashia M, Kasprikova N & Kohoutova M 2014 Heterochromatin variants in human karyotypes: a possible association with reproductive failure. *Reproductive Biomedicine Online* 29 245–250. (https://doi.org/10.1016/j. rbmo.2014.04.021)

© 2022 The authors Published by Bioscientifica Ltd



This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License.

ND aded from Bioscientifica.com at 03/22/2023 11:25:31AM via University of Birmingham and PERPETUAL ACCESS BIRMINGHAM

- Song D, Yin H, Zhang H, Sun Y, Shen C, Du H, Shi M, Sun Z & Sun 2017 Chromosomal polymorphisms do not affect outcome of in vitro fertilization and embryo transfer treatment. *Academic Journal of Second Military Medical University* 38 836–841. (https://doi.org/10.16781/ j.0258-879x.2017.07.0836)
- Wyandt HE, Wilson GN & Tonk VS 2017 Human Chromosome Variation: Heteromorphism, Polymorphism and Pathogenesis. Singapore: Springer.
- Xiaobin Z, Yun F, Qingqing Y, Xiaowei L, Weimin F, Ling W, Zhichao L & Aijun Z 2012 The effect of chromosomal abnormalities on the outcome of in vitro fertilisation and embryo transfer treatment. *Chinese Journal of Andrology* 26 29–32.
- Xu X, Zhang R, Wang W, Liu H, Liu L, Mao B, Zhang X & Zhang X 2016 The effect of chromosomal polymorphisms on the outcomes of

fresh IVF/ICSI-ET cycles in a Chinese population. *Journal of Assisted Reproduction and Genetics* **33** 1481–1486. (https://doi.org/10.1007/s10815-016-0793-2)

Yakin K, Balaban B & Urman B 2005 Is there a possible correlation between chromosomal variants and spermatogenesis? *International Journal of Urology* 12 984–989. (https://doi.org/10.1111/j.1442-2042.2005.01185.x)

Received in final form 4 April 2022 Accepted 14 July 2022 Accepted Manuscript published online 14 July 2022

